Author's response to reviews

Title: Protocol for the Quick Clinical study: a randomised controlled trial to assess the impact of an online evidence retrieval system on decision-making in general practice

Authors:

Enrico Coiera Prof. (e.coiera@unsw.edu.au)
Farah Magrabi Dr. (f.magrabi@unsw.edu.au)
Johanna I Westbrook Prof. (i.westbrook@unsw.edu.au)
Michael R Kidd Prof. (michael@gp.med.usyd.edu.au)
Richard O Day Prof. (r.day@unsw.edu.au)

Version: 3 Date: 20 August 2006

Author's response to reviews:

Editor
BMC Medical Informatics and Decision Making

Authors: Enrico Coiera Prof., Farah Magrabi Dr., Johanna I Westbrook Prof., Michael R Kidd Prof. and Richard O Day Prof.
Title: Protocol for the Quick Clinical study: a randomised controlled trial to assess the impact of an online evidence retrieval system on decision-making in general practice [ISRCTN03597773]
Journal: BMC Medical Informatics and Decision Making
MS: 1491764364108785

Subject: formatting changes

Dear Sir or Madam,

Please find attached a new version of the manuscript. The requested changes have been made as follows:

1. Author list - Can we ask you to link each author to their corresponding affiliation with the use of superscript numbers.
   **Superscripts numbers used**

2. Please insert the section heading Abstract.
   **Inserted**

3. Abstract - Please include a section headed 'Discussions' within your Abstract, remember to update the Submission page.
   **We have included a conclusions section which is more appropriate than a discussions section for a protocol. Submission page has also been updated.**

4. Figure legends - Could you entitle the figure description section 'Figure legends'.
   **Included**

5. Figure 3 - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping.
   **Figure 3 has been cropped as requested.**

We thank you again for the opportunity to publish in your journal.

Yours sincerely
Farah Magrabi